**Author's response to reviews**

**Title:** Spontaneous resolution of chyloous ascites after delivery: A Case Report

**Authors:**

Inas I Babic Dr. (aabinasa@yahoo.com)
Maha I Tulbah Dr. (mahatulbah@kfshrc.edu.sa)
Samir I Ghourab Dr. (samirghourab@kfshrc.edu.sa)

**Version:** 2  **Date:** 2 February 2012

**Author's response to reviews:** see over
As it was advised by the reviewers of the manuscript, I provided some changes in the format of the text.

Native English speaker performed revision too, and corrections were made.

The title was reformed. The gender, sex and ethnicity of the patient were added in the text (underlined words).

I sent separately files with the answers for the reviewers’ concerns.